RESEARCH

Reproductive Health





The influence of quality maternity waiting homes on utilization of facilities for delivery in rural Zambia

Elizabeth G. Henry¹, Katherine Semrau^{2,3,4}, Davidson H. Hamer^{1,5,6}, Taryn Vian¹, Mary Nambao⁷, Kaluba Mataka⁶ and Nancy A. Scott^{1*}

Abstract

Background: Residential accommodation for expectant mothers adjacent to health facilities, known as maternity waiting homes (MWH), is an intervention designed to improve access to skilled deliveries in low-income countries like Zambia where the maternal mortality ratio is estimated at 398 deaths per 100,000 live births. Our study aimed to assess the relationship between MWH quality and the likelihood of facility delivery in Kalomo and Choma Districts in Southern Province, Zambia.

Methods: We systematically assessed and inventoried the functional capacity of all existing MWH using a quantitative facility survey and photographs of the structures. We calculated a composite score and used multivariate regression to quantify MWH quality and its association with the likelihood of facility delivery using household survey data collected on delivery location in Kalomo and Choma Districts from 2011–2013.

Results: MWH were generally in poor condition and composite scores varied widely, with a median score of 28.0 and ranging from 12 to 66 out of a possible 75 points. Of the 17,200 total deliveries captured from 2011–2013 in 40 study catchment area facilities, a higher proportion occurred in facilities where there was either a MWH or the health facility provided space for pregnant waiting mothers compared to those with no accommodations (60.7% versus 55.9%, p < 0.001). After controlling for confounders including implementation of *Saving Mothers Giving Life*, a large-scale maternal health systems strengthening program, among women whose catchment area facilities had an MWH, those women with MWHs in their catchment area that were rated medium or high quality had a 95% increase in the odds of facility delivery than those whose catchment area MWHs were of poor quality (OR: 1.95, 95% Cl 1.76, 2.16).

Conclusions: Improving both the availability and the quality of MWH represents a potentially useful strategy to increasing facility delivery in rural Zambia.

Trial registration: The Zambia Chlorhexidine Application Trial is registered at Clinical Trials.gov (identifier: NCT01241318)

Keywords: Maternity waiting home, Facility-based delivery, Maternal health, Newborn health, Skilled birth attendance, Zambia

* Correspondence: nscott@bu.edu

Full list of author information is available at the end of the article



© The Author(s). 2017 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

¹Department of Global Health, Boston University School of Public Health, Boston, USA

Plain English summary

Most maternal deaths could be averted with improved access to skilled care and facilities equipped to handle obstetric emergencies. In Zambia, where only 56% of rural women deliver in a facility, cost and distance are critical barriers to accessing care. Maternity waiting homes (MWH) have been proposed to address the distance problem. The purpose of this study was to determine if women who have access to MWH are more likely to deliver at a facility, and if the quality of the MWH matters.

Data sources from two separate studies conducted in Southern Province, Zambia, between 2011 and 2013 included survey data from 17,200 pregnant women enrolled at their first antenatal care visit and followed through delivery, and both health facility and MWH assessments. The woman-level data included background characteristics, collected during the enrollment survey, and self-reported location of delivery, collected during a household survey after delivery. The health facility and MWH assessments included indicators of capacity and quality. Statistical methods were used to examine the relationship between utilization of facilities for delivery and MWH quality.

In our study, women whose catchment area health facilities had an MWH or a designed waiting space had higher rates of facility delivery. Moreover, the higher the quality of the MWH, the more likely a woman was to deliver at a facility, regardless of the facility's capacity to address obstetric emergencies. MWH are a potential solution to the distance problem and should be considered as one possible intervention to improve access to facility delivery in Zambia.

Background

An estimated 62% of global maternal deaths occur in sub-Saharan Africa, where a woman's lifetime risk of maternal death is 1 in 59, far higher than the risk in all low-income countries, estimated at 1 in 160 [1]. More than 80% of maternal deaths are due to direct causes associated with obstetric complications [2] and could be prevented with the provision of timely and appropriate intrapartum care. The World Health Organization (WHO) has recommended skilled care at every birth, which includes having a skilled attendant (someone trained to manage normal pregnancies and to identify, manage and refer complications) present for the birth as well as access to facilities with the capacity for emergency obstetric care [3]. In Zambia, the lifetime risk of maternal death is 1 in 38 [1], and just about half of women living in rural areas of the country (56%) deliver at facilities, compared to 89% in urban settings [4]. As with other low-income countries, Zambia continues to face persistent challenges to implementing the WHO skilled birth attendance recommendation.

The factors contributing to the delays in seeking, reaching and accessing quality maternal care, per the Three Delay model, are well established. Key factors affecting utilization and health outcomes include household-level illness recognition and awareness of obstetric complications; women's status, education level, and other socioeconomic factors; perceived accessibility of health care facilities and perceived quality of care that the woman would receive; community-level transportation and referral challenges, and whether a facility actually has adequate infrastructure and clinician capacity to recognize and address clinical needs [5, 6]. Costs, transport and distance to the facility, which lead to the second delay of reaching care, have been repeatedly identified as key drivers of the low utilization of facilities for maternity care in Zambia [7-10]. While the government has abolished user fees for maternal and child services to increase financial access to health care services in 2006, about 65% of the population lives in rural areas and still face physical barriers to access. A recent analysis of the effects of the abolished user-fee policy in Zambia suggests that this reform has not overcome key barriers to utilization of public sector facilities for delivery, and that both quality of care and difficulties related to distance may be a more important determinants [11]. Additional evidence confirms that the odds of facility delivery in rural Zambia decreases as distance to a facility increases [12].

Maternity waiting homes (MWHs), residential lodging near a health facility, represent a potential strategy to improve accessibility and utilization of facilities for delivery. The WHO has recommended MWHs as an intervention to improve maternity care [13]. Women who might otherwise not have access to skilled care due to the constraints posed by distance could benefit by staying at a MWH and being closer to a facility that can manage emergency obstetric complications. MWHs of some sort have been implemented since the beginning of the 20th century in more than 18 countries around the world, including the United States, Canada and Northern Europe, Cuba, India and several countries in sub-Saharan Africa including Zimbabwe, Nigeria, Uganda, Ethiopia and Malawi [14, 15]. However, there is mixed evidence of the effectiveness of MWH on both utilization of health facilities for maternity care and health outcomes [16].

With governments eager to find feasible solutions to the distance problem, there is a resurgence of interest in constructing MWH and developing policies around their implementation. However, it is not yet clear if this will be a worthwhile investment among competing priorities. Evidence from Zimbabwe suggests that when a facility has an MWH women are much more likely to deliver there [17]. The perceived quality of the MWH, among other factors such as direct and indirect costs of staying at the MWH, may also play a role in rates of MWH utilization, including in rural Zambia [15, 18]. However, there is no evidence on whether or not the actual quality of the MWH is associated with utilization of facilities for delivery.

In Zambia, women are more likely to deliver in facilities better equipped to handle obstetric emergencies, independent of distance to the facility, suggesting that quality of care is also an important factor in decision making about facility-based birth [12]. It is important, therefore, to also understand whether the relationship between MWH quality and facility utilization would remain regardless of the quality of the health facility in terms of capacity to deal with obstetric emergencies. It is possible that the quality of the health facility is the key driver of facility delivery, rather than the quality of the MWH. The purpose of our study was two-fold: to determine whether the existence of an MWH at a facility would predict women's utilization of the facility for delivery, and to determine whether the quality level of the MWH would predict the magnitude of this relationship, adjusting for facility quality to handle obstetric emergencies.

Methods

Study setting

Our study utilized data collected from two separate studies conducted in two contiguous districts in Southern Province, Kalomo and Choma Districts, between 2011 and 2013: 1) a formative evaluation of MWH conducted in 2013 on the physical quality of MWHs [19]; and 2) a cluster-randomized controlled trial conducted in Southern Province between 2011 and 2013 [20].

At the time of data collection, Kalomo had a primarily rural population (93%) of 258,570 [21] and 35 health facilities, including 27 health centers (HC), six health posts and two referral hospitals [22]. Choma District also had a mostly rural (76%) population of 247,860 [21] with 29 HCs, six health posts and two referral hospitals. We would most likely expect to find an MWH at the HCs and hospitals, but not health posts, as they do not provide delivery services. Of those hospitals and HCs that offer delivery services, at the time of our survey 25 of Kalomo facilities had MWHs (86%), whereas only six in Choma had MWHs (19%) [22].

In Kalomo District during the time of our study, there was an ongoing initiative to improve maternal health by addressing the three delays through a public-private partnership called *Saving Mothers, Giving Life (SMGL)* [23–25]. The package of interventions implemented through SMGL since 2012 have included, among others, community mobilization and sensitization activities, improvements in referral systems, mentoring health staff, and investments in supply chain and equipment at facilities. There is evidence that SMGL had a significant impact on rates of facility delivery in Kalomo [26]. Therefore, the effect of this package of interventions is addressed in our analysis as a potential confounder.

Data sources

Data on facility delivery were obtained from the Zambia Chlorhexidine Application Trial (ZamCAT), a cluster Randomized Controlled Trial (RCT) in which 39,797 pregnant women in six districts of Southern Province, including Choma and Kalomo, were enrolled at their first antenatal care (ANC) visit and followed through 28 days post-delivery [20, 27]. The goal of ZamCAT was to evaluate the effectiveness of using chlorhexidine cord cleansing to reduce neonatal mortality. The woman-level data included background characteristics, collected during the initial enrollment survey, and location of their delivery, collected during a survey after delivery at the 1 and 4 day postpartum household visits.

Data for facility capacity for emergency obstetric care were obtained from a health facility assessment (HFA) tool conducted as part of ZamCAT between June and August 2013. Not all facilities in each of the two districts were included in the ZamCAT study. Facility-level criteria for inclusion in the study were: (1) an estimated 160 births per year in the catchment area, (2) routine provision of ANC services, and (3) willingness to participate. In Kalomo District, 22 facilities, all HC, were selected, representing 81% of all HC in the district and in Choma 18 facilities (HC) were selected, representing 62% of all HC in the district. The HFA tool captured basic indictors of capacity to perform maternal and newborn health signal functions and other indicators of routine maternity and newborn care. Signal functions are a set of medical interventions that address the direct causes of maternal death [28]. Full details of the ZamCAT trial are described elsewhere [20, 27].

Indicators

In this analysis, the main outcome indicator was delivery at any facility, determined by location of birth (health facility or hospital) reported by the mother at the Zam-CAT household postpartum visit. Indicators of woman's socio-economic status (household wealth, education level), maternal demographics (age, marital status), and pregnancy characteristics (parity, ANC) were obtained from the ZamCAT enrollment questionnaire.

The indicator for facility capacity for emergency obstetric care was a continuous score, calculated as the sum of the basic emergency obstetric and newborn care (EmONC) signal functions (maximum of 7) [28] and 1 point for each of the following: electricity, water, 24-h care, and availability of a skilled provider (defined by the WHO as someone trained to manage normal pregnancies and to identify, manage and refer complications) [3], for a total 11 possible points. Signal functions were assessed by asking the facility in-charge interviewee whether or not the function had been performed within the last 3 months at that facility.

The primary independent variable was a composite quality score for the MWH. As part of a formative evaluation [19], the study team systematically assessed and inventoried the functional capacity of all existing MWHs in the selected districts (n = 31; 25 in Kalomo and 6 in Choma). The health facilities with existing MWHs were identified in advance by the district health offices. The composite quality score was created by utilizing both quantitative and qualitative data collected at each MWH, inclusive of a series of questions asked of a member of affiliated clinic staff as well as photos that captured the state of the physical structure/space, availability of a water source within 200 m, availability/state of a toilet and bedding, and availability/state of a gathering and cooking area. These items emerged from our literature review of the physical factors that may be important determinants of quality in an MWH. In seven cases, facilities did not have a separate MWH structure at the time of assessment, but instead utilized a clinic area as a designated space for waiting women. We evaluated these spaces using the same criteria.

The scoring system for each criterion ranged from 0 (not available) to 5 (present and fully functional). There were a total possible 75 points if all criteria for each of the 15 components was rated highest. Based on their composite score, MWHs were also categorized into tertiles and labeled as "low", "medium" and "high" quality.

Analysis

We limited our analysis to those catchment areas for which we had health facility and woman-level data from ZamCAT (n = 40 sites; 18 of these had an affiliated MWH, three had a designated area for waiting mothers, often the clinic wards, and 19 of these had no structure nor designated area). Our sample of sites excluded seven health facilities in Kalomo and three in Choma that had an MWH but for which we did not have ZamCAT woman-level data. Of these 10 excluded, three facilities were hospitals, as hospital catchment areas defined differently than for HCs, and thus were not assigned for randomization for ZamCAT. The other non-hospital health facilities/HCs were not included in the ZamCAT study because they did not meet the health facility inclusion criteria.

We conducted bivariate analyses examining associations between background characteristics of the women in our sample and our outcome of facility delivery to determine what factors to control for in the regression analyses. We used the Pearson chi-square test for categorical variables and t-tests for continuous variables if the data were normally distributed or non-parametric Wilcoxon rank-sum tests if non-normally distributed. Any characteristics associated with outcome variables with *p*-value <0.20 were included in the adjusted logistic regression model.

We used multiple logistic regression to assess the likelihood of facility delivery based first on the facility capacity score and then on the composite MWH quality score. We also regressed the primary individual-level outcome (facility delivery) against the category of the MWH quality (low, medium, high), adjusting for covariates that may have moderated the effect, such as sociodemographic characteristics and distance to the facility, as well as for facility capacity score and level of SMGL program implementation. SMGL level of implementation was defined by using three time periods: data on women who 1) delivered before or during January 2012, the time at which the SMGL rollout started; 2) women who delivered between February and August 2012 and may have had some exposure to the SMGL program; and 3) women who delivered from September 2012 to the end of the ZamCAT data collection in October 2013 and were most likely exposed to some level of SMGL activities. In Choma the SMGL implementation level was always zero as SMGL never operated in that district during those time periods. Quantitative data were analysed in SAS version 9.3 [29].

Results

Sample characteristics

A total of 17,200 women were included in the final analysis. Women had a median age 25.0 (IQR 20.0-31.0), 85.4% were married and more than 91% had at least some primary education (Table 1). Just over 22% had no children prior to this pregnancy, and 37% reported living more than two hours from a health facility (regardless of type of transportation used). Just over 5% of women were recorded as HIV positive on their ANC cards. More than half of women in our sample (58.5%) delivered at any facility (n = 10,069). Woman's age, marital status, education level, parity, ANC visits, HIV status and distance to facility were all associated with facility delivery. Older women, women who were married, had fewer years of education, more than one child, lived more than two hours away from a health facility, had less than the four recommended ANC visits, and were not HIV positive were less likely to deliver in a facility. Just over half of women (55%) lived in catchment areas with an existing MWH or designated waiting space for pregnant mothers.

Characteristics of health facilities, both those with and without affiliated MWHs, as well as those that accommodate pregnant women by designating some space for waiting, are included in Table 2. On the 11-point scale for the composite score, facility scores ranged from 2 points to 9 points with a mean of 5.8 (SD 1.8). Facilities with an MWH, as well as those that provide space for waiting pregnant women,

Characteristic	Sample distribution $n = 17,200$ (%)	Delivered in a facility $N = 10,069$ (%)	Delivered at home N = 7131 (%)	Unadjusted odds of delivering in a facility
Woman's age in years (%)				
15–19	22.4	25.6	17.8	1.0
20–24	27.5	26.6	28.8	0.64 (0.59, 0.70)***
25–34	36.3	34.6	38.8	0.62 (0.57, 0.67)***
35–49	13.8	13.3	14.5	0.64 (0.57, 0.71)***
Age in years (median, IQR)	25.0 (20.0-31.0)	24.0 (19.0-30.0)	25.0 (21.0-31.0)	
Household size (median, IQR)	6.0 (4.0-8.0)	6.0 (4-8)	6.0 (4–8)	
Currently married (%)				
No	14.6	17.4	10.6	1.0
Yes	85.4	82.6	89.4	0.56 (0.51, 0.62)***
Mother's highest education (%)				
None (0)	8.3	7.0	10.2	1.0
Any primary (1–7)	55.0	50.0	62.1	1.17 (1.05, 1.31)**
More than primary (7+)	36.7	43.0	27.7	2.27 (2.02, 2.55)***
Asset quartile (%)				
First/poorest	26.6	25.7	27.7	1.0
Second	24.5	22.9	26.9	0.92 (0.84, 0.99)*
Third	24.7	24.8	24.6	1.09 (0.99, 1.18)
Fourth/highest	24.2	26.6	20.8	1.38 (1.27, 1.51)***
Parity (%)				
0	22.3	27.7	15.6	1.0
1	17.7	17.5	18.1	0.51 (0.46, 0.56)***
>=2	60.0	54.8	67.4	0.43 (0.39, 0.46)***
Distance to health facility \geq 2 h (%	o)** ^a			
No	62.7	69.1	53.6	1.0
Yes	37.3	30.9	46.4	0.52 (0.49, 0.55)***
4 ANC visits (%)				
No	55.0	48.7	63.9	1.0
Yes	45.0	51.3	36.1	1.87 (1.75, 1.99)***
Mother HIV positive (%)				
No	94.4	92.8	96.8	1.0
Yes	5.6	7.2	3.2	2.34 (2.00, 2.74)***

 Table 1 Characteristics of the women with and without facility delivery and unadjusted odds of facility delivery

^aSelf-reported time from home to nearest health facility

*p <0.05, **p <0.01, ***p <0.001

appeared to differ systematically from those that did not have a MWH, with those with an MWH having a higher overall score of 6.8 (SD 1.5) compared to those without (4.8 (SD 1.6)). Nearly 60% of facilities with an MWH had performed four or more signal functions, compared to only 16% of those without a MWH. A much higher proportion of facilities with a MWH had administered antibiotics for maternal infections in the past three months (100.0% versus 42.1%). This trend was the same for nearly all signal functions, though not always as pronounced. None of the facilities without a MWH had electricity, whereas seven facilities with a MWH did. Nearly all facilities had access to clean water, provided obstetric services 24/7 and, except for two facilities with a MWH, had at least one skilled provider trained in deliveries on staff. On average those facilities with an MWH had a higher average number of deliveries per month (22.4 (SD 17.2)) than in those without an MWH (17.2 (SD 15.0)). Facilities with a designated waiting space had the highest average number of deliveries, with a mean of 25.0 (SD 14.0).

Table 2 Characteristics of health facilities, with and without Maternity Waiting Homes (MWH)^a

Facility characteristics	All study facilities	Facilities have an MWH		
	N(%)	None n(%)	Yes n (%)	Designated Area for Waiting Mothers n (%)
	40 (100.0)	19 (100.0)	18 (100.0)	3 (100.0)
Signal functions				
Four or more signal functions	15 (37.5)	3 (15.8)	11 (61.1)	1 (33.3)
Parenteral antibiotics for maternal infection	27 (67.5)	8 (42.1)	18 (100.0)	1 (33.3)
Parenteral oxytocic drugs for hemorrhage	38 (95.0)	18 (94.7)	17 (94.4)	3 (100.0)
Parenteral magnesium sulfate for eclampsia	9 (22.5)	3 (15.7)	4 (22.2)	2 (96.7)
Manual removal of the placenta	10 (25.0)	3 (15.7)	6 (33.3)	1 (33.3)
Removal of retained products of conception	10 (25.0)	2 (10.5)	7 (38.9)	1 (33.3)
Assisted vaginal delivery	4 (10.0)	0 (0.0)	4 (22.2)	0 (0.0)
Resuscitation with bag and mask of non-breathing baby	24 (60.0)	7 (36.8)	15 (83.3)	2 (66.7)
Infrastructure and human resources				
Electricity	8 (20.0)	0 (0.00)	7 (38.9)	1 (33.3)
Water supply	37 (92.5)	17 (89.5)	17 (94.4)	3 (100.0)
Service availability 24/7	36 (95.0)	15 (79.0)	18 (100.0)	3 (100.0)
At least 1 skilled provider	38 (95.0)	19 (100.0)	16 (88.9)	3 (100.0)
Total composite score				
Mean (SD)	5.8 (1.8)	4.8 (1.6)	6.8 (1.5)	6.3 (2.5)
Median (Range)	6.0 (2.0-9.0)	5.0 (2.0-8.0)	7.0 (4.0-9.0)	6.0 (4.0-9.0)
Average number of deliveries (monthly)	20.1 (15.9)	17.2 (15.0)	22.4 (17.2)	25.0 (14.0)

^aThose with no MWH but with designated areas for waiting mothers are presented separately because they are distinctly different from those with a MWH and from those with no MWH

MWHs in our study were generally in poor condition. All MWHs had some type of water supply, toilet, and sanitation inside and outside the shelter. Nearly all of the MWHs had some type of floor, roof and general structural integrity (Table 3). However, most did not have beds or mattresses, nor separate bathing areas for women. We observed wide variation in quality between MWHs, with composite scores ranging from 12 to 66 out of a possible 75, and a median of 28.5 with an inter-quartile range (IQR) of 23–41.

Relationship between maternity home quality and facility delivery

The rate of facility delivery for women whose catchment area had any MWH (including a designated space for waiting) was 60.7 (5698/9385), compared with 55.9 (4371/7815) for those without any MWH or waiting space, an absolute difference of 4.7 points. Adjusting for maternal age, any maternal education, household asset quartile, parity, maternal HIV status, four or more ANC visits, distance to health facility, and whether or not the SMGL program was operational in the district when the mother went for delivery (but not facility capacity for emergency response), there was a 37% relative increase in the odds of facility delivery for women living in catchment areas with an MWH or designated space,

Table 3 Characteristics of 18 existing Maternity Waiting Homes(WH), Kalomo and Choma Districts

MWH characteristics	Number (%) Any present <i>N</i> = 18	Score Mean (SD)	Score Median (IQR)
Beds	2 (11.1)	0.6 (1.6)	0.0 (0–0)
Mattresses	3 (16.7)	0.7 (1.7)	0.0 (0-0)
Electricity	6 (33.3)	1.4 (2.1)	0.0 (0-4)
Sanitation inside	18 (100.0)	3.1 (1.1)	3.0 (2–4)
Sanitation outside	18 (100.0)	2.8 (1.1)	2.5 (2–3)
Security	18 (100.0)	3.1 (1.6)	3.0 (1-5)
Water supply	18 (100.0)	2.9 (0.8)	3.0 (2–3)
Toilet	18 (100.0)	2.6 (0.9)	3.0 (2–3)
Community support	6 (33.3)	0.7 (1.1)	0.0 (0-2)
Structure integrity	18 (100.0)	3.2 (1.2)	3.0 (2–4)
Roof	18 (100.0)	1.0 (0)	1.0 (2–5)
Floor surface	18 (100.0)	4.2 (1.5)	5.0 (2–5)
Shower/bathing area	5 (27.8)	0.7 (1.2)	0.0 (0-0)
Cooking area	18 (100.0)	2.2 (1.3)	2.0 (1-3)
Total score	NA	32.7 (13.9)	28.5 (23–41)

compared to women whose catchment areas did not have any MWH (OR: 1.37, 95% CI: 1.27, 1.46) (results not shown in table). Additionally, we tested for any effect the ZamCAT intervention may have had on facility delivery rates. The unadjusted relationship between women living in ZamCAT intervention areas, as compared to non-intervention areas, and facility delivery was non-significant (p = 0.08, OR: 0.947, 95% CI: 0.89, 1.01) (results not shown in table) and therefore was not included in the adjusted model.

Our results also indicate that facility capacity for emergency response has an independent effect on utilization of facilities for delivery. As quality of the facility increases, women are more likely to delivery there (1.10, 95% CI: 1.08, 1.12) (results not shown). This relationship is still significant when we adjust for confounders (maternal age, any maternal education, household asset quartile, parity, maternal HIV status, four or more ANC visits, distance to health facility, and whether or not the SMGL program was operational in the district when the mother went for delivery) (1.10, 95% CI: 1.08, 1.12). When we then revisited the relationship between having an MWH (or designated space) and facility delivery but adjusted for facility capacity, as well as the covariates listed above, the odds of facility delivery was still 19% higher for those with any MWH or accommodation than those without (1.19, 95% CI: 1.10, 1.29) (Table 4).

When we differentiate between facilities that had an actual MWH structure and those that had only designated accommodations for pregnant waiting mothers, such as space in the wards, as compared to those without any place for waiting, there is a less pronounced positive relationship between facility delivery and MWH among those with just a structure (OR: 1.09, 95% CI:1.03, 1.16), but a large increase in the odds of facility delivery among those women whose catchment areas

provided space but no formal MWH structure (OR:2.47, 95% CI: 2.17, 2.81). Adjusting for the same covariates as above, the odds of facility delivery becomes non-significant for those with just an MWH, compared to those with none (1.07, 95% CI: 0.98, 1.16) but the effect is still positive for those with a designated space (OR: 2.13, 95% CI: 1.85, 2.46).

When we examined the level of quality of the MWH, only among those that were scored (excluding those with waiting spaces as they did not have the same criteria upon which they could be measured), the rate of facility delivery in those with a low quality MWH was 49.3%. This was lower than the rate of facility delivery in catchment areas with a medium or high-level quality MWH (63.8%). When we examined the relationship between MWH quality and facility delivery, again limiting our analysis only to those facilities with existing MWH structures and controlling for those covariates listed above, including facility capacity score, there was a 95% relative increase in the odds of facility delivery for high or medium-ranking MWHs compared to low-ranking shelters, adjusting for covariates (OR: 1.95, 95% CI: 1.76,2.16).

Discussion

Women in our study were more likely to deliver at health facilities when their catchment area health facility had an MWH or a designated space for waiting on the premises than those women with no immediate access to an MWH or waiting space. When controlling for facility capacity to handle obstetric emergencies, the relationship between presence of an MWH or designated space for waiting and utilization of facility for delivery remained. This is consistent with findings from previous work in Zambia that showed that the absence of a suitable shelter for expecting mothers was a contributing factor to low utilization of maternal health services [30]

Table 4 Likelihood of facil	ty-based birth by Materni	ty Home: any and qu	ality level
-----------------------------	---------------------------	---------------------	-------------

	No. of Births	Facility-based Births No (%)	Odds ratio (95% CI)	Adjusted OR (95% Cl)
MWH or Waiting Space at F	acility			
No ^a	7815	4371 (55.9)	1.00	1.00
Yes	9385	5698 (60.7)	1.22 (1.15, 1.29)	1.19 (1.10, 1.29)
Type of Waiting Space				
None ^a	7815	4371 (55.9)	1.00	1.00
Space but no MWH	1392	1055 (75.8)	2.47 (2.17, 2.81)	2.13 (1.85, 2.46)
MWH	7993	4643 (58.1)	1.09 (1.03, 1.16)	1.07 (0.98, 1.16)
Rank of MWH ^c				
Low ^a	3142	1550 (49.3)	1.00	1.00
Medium/High	4851	3093 (63.8)	1.81 (1.65, 1.98)	1.95 (1.76, 2.16)

^areference group

^bControlling for maternal age, any maternal education, asset quartile, parity, maternal HIV status, four or more ANC visits, distance to health facility, SMGL program implementation, facility capacity

^cOnly among those with an existing MWH structure and a score (missing 3 facilities with designated spaces that were not able to be scored)

and a study in Liberia where availability of MWHs decreased the distance barrier for women to access skilled care for childbirth [31].

The finding that those women with a designated space had a much higher odds of facility delivery than those with no MWH or space suggests that there may be a characteristics about the facility, such as perceived supportive staff that are willing to accommodate waiting women when no space is formally available, that may influence women's decisions to deliver there. Women's cultural beliefs and perceptions of facility-based care have been found to be major drivers of facility delivery [32]. In Tanzania, a woman's previous or current experience with the health system, including whether or not she was treated with respect during ANC or delivery, was found to be more important that the perceived quality of the facility itself [33]. Our study findings indicate there is something unique about facilities that provide waiting space for mothers, and this warrants further investigation.

In our study the quality of the MWH is an important determinant of utilization. Among those facilities that had an MWH structure, women were more likely to deliver at facilities if the structures were of higher quality. The low-quality MWH had about the same utilization rates as those with no MWH at all. This suggests that if MWH are constructed in the future, particular attention should be paid to both the inclusion and the quality of components such as proper bedding, private spaces for bathing, and areas for cooking, among others. These findings are consistent with results from a review of MWH utilization which cited several factors that influence a woman's decision to stay in an MWH, including the small size of the home and the lack of proper hygiene of the facilities [16]. In our region of Zambia, the poor state of the MWHs, including inadequate sleeping spaces and bedding, water and sanitary services, have been cited as factors that deterred women from utilizing MWH [18]. Additional research from Zambia has illuminated that poor conditions of the MWH led some husbands to forbid their wives from using them [34]. Community acceptance and support have been shown to be critical factors in the successful introduction of MWHs in Ethiopia [35]. Our findings provide additional evidence that considering the needs of the community will be integral to the potential success of the utilization of MWH in Zambia.

There are several limitations to this analysis. First, although we have the self-reported type of facility at which a woman delivered, we were not able to determine the actual delivery location of the woman, whether at her health facility catchment area (HFCA), or a neighboring facility or hospital. Our study is assuming that most women delivered in their catchment area's facility, and would have been influenced by the presence or absence of an MWH in their HFCA. It is possible that women delivered outside of their catchment area, and were influenced by the state of their HFCA's MWH and/or that of a neighboring HFCA.

It is important to note that our data only indicate an association between quality of MWH and utilization of facilities for delivery. We do not know if an MWH is scored as higher quality because it was improved due to an increase of facility deliveries at that site, or if the volume of deliveries changed as a result of the introduction of or improved quality of an MWH. It is also possible that there were substantial changes to the condition of the MWH and/or the facility between the time period for which we captured data on women's deliveries. It is possible that when the MWH were assessed in 2013, they had deteriorated significantly since 2011, and therefore we are overestimating the rate of facility delivery at low-quality shelters that had previously been medium or high quality shelters. However, this would bias results towards the null and therefore our estimates are likely to be conservative. Another study (Henry et al., under review) suggests that there was very little change in the health facilities' capacity for EmONC between 2011 and 2013 in Kalomo District during SMGL implementation. We also controlled for SMGL implementation, which would have likely accounted for any improvements at the facilities and the MWH in Kalomo during that time, as well as significant investments in the training and support of Safe Motherhood Actions Groups (SMAGs), community health volunteers who promote safe delivery and birth practices at the household level.

There are additional deterrents to MWH utilization that we were not able to capture in our study, including an increase in costs of delivery at a facility compared to delivery at home in MWHs, lack of privacy, and lack of respect from health staff [16]. Other factors may be operating in women's decision making about utilizing facilities for delivery, such as attitudes, personal norms, and behavioral control [7, 36]. The lack of either provision of food or help with cooking and the limited availability of water and firewood for cooking, were among other factors related to quality that affected a woman's staying in an MWH in both the global review of MWH utilization as well as the study in Zambia [16, 18]. We were not able to systematically assess these factors in our study, though they may also have had an influence. It also not clear whether or not there are unmeasurable characteristics about the communities that have MWHs that contribute to women's choosing to deliver at a facility, such as a supportive community environment for maternal health, community outreach activities independent of SMGL, or community chiefs and leadership promoting facility delivery more than in other areas.

Lastly, the generalizability of the findings may be limited, as data were collected from only two districts in Southern Province. Findings are perhaps most generalizable to other primarily rural districts that are culturally or geographically similar to the two study sites.

Conclusions

Our study provides evidence to suggest that MWHs could be a feasible option to increasing access to facility-based delivery in this context by addressing some of the factors leading to the first and second delays in seeking and reaching maternity care, and that implementation of MWHs in Zambia should pay close attention to community-accepted quality measures. However, it is important to note that MWHs are a link between women in communities and the formal health system, and that MWHs alone would not be sufficient to improve health outcomes for mothers and babies. Other components critical to maternal and newborn care must also be available within the health system including: a definition of risk factors and selection criteria for mothers; a system for identification and referral of the woman; availability of basic and comprehensive EmONC services and providers, and community support [14]. Constructing even high quality MWHs without consideration of these other factors could create demand for quality maternity services without adequate supply, exacerbating factors that lead to the third delay. It will be important to assess changes in service utilization and maternal and newborn health outcomes as further anticipated investments are made into the construction and implementation of MWH throughout Zambia.

Abbreviations

ANC: Antenatal care; EmONC: Emergency obstetric and newborn care; HC: Health center; HFA: Health facility assessment; IQR: Interquartile range; MWH: Maternity waiting home; SD: Standard deviation; SMAG: Safe Motherhood action Groups; SMGL: Saving Mothers, Giving Life; WHO: World Health Organization; ZamCAT: Zambia chlorhexidine application trial

Acknowledgements

The authors would like to thank and acknowledge the work of the ZamCAT study team, the Maternity Waiting Homes Formative Evaluation study team, SMGL program staff in Zambia, and the Ministry of Health, District and Provincial Medical Offices and Chiefs of Southern Province. We especially thanks those who worked with us on the collection of this data including Rachel Fong, Jeanette Kaiser, Thandiwe Ngoma, Sara Gille, as well as RTC Fellows Pam Polino and Jaron Link. We also would like to express our gratitude to the women, their newborns and their families who participated in ZamCAT and to the health facility staff who participated in the MWH formative study. The activities discussed in the manuscript were partially supported by funding from Merck through its Merck for Mothers program and the RTC Fellowship. The content of this manuscript is solely the responsibility of the authors and does not represent the official views of Merck. This manuscript is based partially on data from a study funded by the Bill & Melinda Gates Foundation. The findings and conclusions contained within are those of the authors and do not necessarily reflect positions or policies of the Bill & Melinda Gates Foundation.

Funding

This program was funded by and developed and is being implemented in collaboration with Merck for Mothers, Merck's 10-year, \$500 million initiative to help create a world where no woman dies giving life. Merck for Mothers is known as MSD for Mothers outside the United States and Canada, the MSD Fellowship for Global Health and the Bill & Melinda Gates Foundation.

Availability of data and materials

The datasets analyzed during the current study are not publicly available due to IRB constraints but are available from the corresponding author on reasonable request.

Authors' contributions

NS designed and led the data collection and primary analysis for the formative study. KS and DH designed and implemented ZamCAT. EH and NS designed and conducted the analysis for the current study and drafted the manuscript. NS, KS and DH made substantial contributions to the study design, interpretation of results and critical revision of the manuscript. TV, MN and KM contributed to the ongoing review of the study results and critical revision of the final draft. All authors have approved the final version.

Competing interests

The authors declare that they have no competing interests.

Consent for publication

Not applicable.

Ethics approval and consent to participate

Ethical approvals were obtained from the Boston University Institutional Review Board (H-32470 and H-29647) and the ERES Converge Ethics Committee in Zambia (Ref.No.2013-Aug-008) for the formative study and the University of Zambia Research Ethics Committee (ID 016-08-10) for ZamCAT. Both studies were approved by the Ministry of Health Research Secretariat and the Southern Province Medical Office. The informed consent process was used for all study participants in both studies.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Department of Global Health, Boston University School of Public Health, Boston, USA. ²Ariadne Labs, Boston, USA. ³Division of Global Health Equity, Brigham & Women's Hospital, Boston, USA. ⁴Department of Medicine, Harvard Medical School, Boston, USA. ⁵Section of Infectious Diseases, Department of Medicine, Boston University School of Medicine, Boston, USA. ⁶Zambia Center for Applied Health Research and Development (ZCAHRD) Limited, Lusaka, Zambia.

Received: 12 February 2017 Accepted: 19 May 2017 Published online: 30 May 2017

References

- World Health Organization (WHO). Trends in Maternal Mortality: 1990 to 2013. Estimates by WHO, UNICEF, UNFPA, The World Bank and the United Nations Population Division. 2014
- Safe Motherhood Programme, World Health Organization (WHO). Mother-Baby Package: Implementing safe motherhood in countries. Geneva; 1994.
- World Health Organization (WHO). Making pregnancy safer: the critical role of the skilled attendant a joint statement by WHO, ICM and FIGO. 2004.
- Central Statistical Office (CSO) Zambia, [Ministry of Health (MOH) [Zambia], ICF International. Zambia Demographic and Health Survey 2013–14. Rockville, Maryland, USA; 2014.
- Gabrysch S, Campbell OMR. Still too far to walk: literature review of the determinants of delivery service use. BMC Pregnancy Childbirth. 2009;9:34.
- Thaddeus S, Maine D. Too far to walk: maternal mortality in context. Soc Sci Med. 1994;38:1091–110.
- Sialubanje C, Massar K, Hamer DH, Ruiter RAC. Personal and environmental predictors of the intention to use maternal healthcare services in Kalomo, Zambia. Health Educ. Res. 2014

- Lohela TJ, Campbell OMR, Gabrysch S. Distance to Care, Facility Delivery and Early Neonatal Mortality in Malawi and Zambia. PLoS One. 2012;7.
- Kyei NNA, Campbell OMR, Gabrysch S. The influence of distance and level of service provision on antenatal care use in rural Zambia. PLoS One. 2012;7:e46475.
- Sialubanje C, Massar K, Hamer DH, Ruiter RAC. Reasons for home delivery and use of traditional birth attendants in rural Zambia: a qualitative study. BMC Pregnancy and Childbirth. 2015;15:216.
- Chama-chiliba CM, Koch SF. An assessment of the effect of user fee policy reform on facility-based deliveries in rural Zambia. BMC Res. Notes. BioMed Central; 2014;29.
- Gabrysch S, Cousens S, Cox J, Campbell OMRO. The influence of distance and level of care on delivery place in rural Zambia: a study of linked national data in a geographic information system. PLoS Med. 2011;8:e1000394.
- World Health Organization. WHO recommendations on health promotion interventions for maternal and newborn health. 2015. p. 94.
- 14. WHO. Maternity Waiting Homes: A review of experiences. Geneva WHO. 1996;96:1–44.
- Gaym A, Pearson L, Soe KWW. Maternity waiting homes in Ethiopia–three decades experience. Ethiop Med J. 2012;50:209–19.
- van Lonkhuijzen L, Stekelenburg J, van Roosmalen J. Maternity waiting facilities for improving maternal and neonatal outcome in low-resource countries. Cochrane Database Syst Rev. 2012;10:CD006759.
- Van Den Heuvel OA, de Mey WG, Buddingh H, Bots ML. Use of maternal care in a rural area of Zimbabwe: a population-based study. Acta Obstet Gynecol Scand. 1999;78:838–46.
- Sialubanje C, Massar K, G van der Pijl MS, Maria Kirch E, Hamer DH, C Ruiter RA. Improving access to skilled facility-based delivery services: Women's beliefs on facilitators and barriers to the utilisation of maternity waiting homes in rural Zambia. Reprod. Health. 2015
- Scott N, Biemba G, Hamer DH, Vian T, Semrau K, Simon JL, et al. Improving equitable access to safe deliveries in Zambia: Findings from a formative evaluation. Boston, MA; 2014.
- Semrau KEA, Herlihy JM, Grogan C, Musokotwane K, Yeboah-Antwi K, Mbewe R, et al. Effectiveness of 4% chlorhexidine umbilical cord care on neonatal mortality in Southern Province, Zambia: a cluster randomised controlled trial. Lancet Glob. Heal. 2016
- 21. Central Statistical Office (CSO) Zambia, Road N. 2010 census of population and housing. 2012.
- 22. Republic of Zambia Minist. List of health facilities in Zambia. 2011. p. 1–115.
- 23. Saving Mothers Giving Life. Saving mothers, giving life making pregnancy & childbirth safer in Uganda and Zambia: annual report 2013. 2014. p. 1–2.
- Kruk ME, Rabkin M, Grépin KA, Austin-Evelyn K, Greeson D, Masvawure TB, et al. "Big push" to reduce maternal mortality in Uganda and Zambia enhanced health systems but lacked a sustainability plan. Health Aff. 2014; 33:1058–66.
- Kruk ME, Vail D, Austin-Evelyn K, Atuyambe L, Greeson D, Grepin KA, et al. Evaluation of a maternal health program in Uganda and Zambia finds mixed results on quality of care and satisfaction. Health Aff. 2016;35:510–9.
- Henry EG, Thea DM, Hamer DH, DeJong W, Musokotwane K, Chibwe KC, et al. The impact of a multi-level maternal health program on facility delivery and capacity for emergency obstetric care in Zambia. Under Rev. 2017;
- Hamer DH, Mazimba A, Musokotwane K, Banda B, Mpamba C, Mwangelwa B, et al. Engagement of the community, traditional leaders, and public health system in the design and implementation of a large communitybased cluster randomized trial of umbilical cord care in Zambia. :1–29.
- World Health Organization (WHO), UNFPA, UNICEF, AMDD. Monitoring emergency obstetric care. In: A handbook. 2009.
- SAS Institute. SAS, version 9.3 of the SAS system for windows. In: Cary, NC: SAS Institute Inc. Product or service names are registered trademarks or trademarks of SAS institute Inc., Cary, NC, USA. 2011.
- Stekelenburg J, Kyanamina S, Mukelabai M, Wolffers I, van Roosmalen J. Waiting too long: low use of maternal health services in Kalabo, Zambia. Trop Med Int Health. 2004;9:390–8.
- Lori JR, Wadsworth AC, Munro ML, Rominski S. Promoting access: the use of maternity waiting homes to achieve safe motherhood. Midwifery. 2013;29:1095–102.
- Moyer CA, Dako-Gyeke P, Adanu RM. Facility-based delivery and maternal and early neonatal mortality in sub-Saharan Africa: a regional review of the literature. Afr J Reprod Health. 2013;17:30–43.

- Larson E, Hermosilla S, Kimweri A, Mbaruku GM, Kruk ME. Determinants of perceived quality of obstetric care in rural Tanzania: a cross-sectional study. BMC Health Serv Res. 2014;14:483.
- Sialubanje C, Massar K, Kirch EM, van der Pijl MSG, Hamer DH, Ruiter RAC. Husbands' experiences and perceptions regarding the use of maternity waiting homes in rural Zambia. Int J Gynecol Obstet Int Fed Gynecol Obstet. 2016;133:108–11.
- Kelly J, Kohls E, Poovan P, Schiffer R, Redito A, Winter H, et al. The role of a maternity waiting area (MWA) in reducing maternal mortality and stillbirths in high-risk women in rural Ethiopia. BJOG. 2010;117:1377–83.
- Sialubanje C, Massar K, Hamer DH, Ruiter RAC. Understanding the psychosocial and environmental factors and barriers affecting utilization of maternal healthcare services in Kalomo, Zambia: a qualitative study. Health Educ Res. 2014;29:521–32.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

